Author's response to reviews

Title: Improving the outcomes of primary care attenders with common mental disorders in developing countries: a cluster randomized controlled trial of a collaborative stepped care intervention in Goa, India

Authors:

Vikram H Patel (vikram.patel@lshtm.ac.uk)
Betty R Kirkwood (betty.kirkwood@lshtm.ac.uk)
Sulochana Pednekar (sulochana@sangath.com)
Ricardo Araya (r.araya@bristol.ac.uk)
Michael King (mking@medsch.ucl.ac.uk)
Daniel Chisholm (chisholmd@who.int)
Gregory Simon (simon.g@ghc.org)
Helen Weiss (helen.weiss@lshtm.ac.uk)

Version: 5 Date: 18 January 2008

Author's response to reviews:

Dear editors

I have revised the paper to conform to all the formatting revisions requested in your email dated January 17th. Thus, I am now uploading two files, one with the manuscript and one with the figure. I am afraid I was unable to reformat the figure to get any smaller than it is currently.

Vikram